Author’s response to reviews

Title: Asian-White Differences in Short Sleep Duration by Industry of Employment and Occupation in the US: A Cross-sectional Study

Authors:

Chandra L Jackson (cjackson@hsph.harvard.edu)
Ichiro Kawachi (ikawachi@hsph.harvard.edu)
Susan Redline (sredline1@rics.bwh.harvard.edu)
Hee-Soon Juon (hjuon@jhsph.edu)
Frank B Hu (fhu@hsph.harvard.edu)

Version: 2 Date: 10 December 2013

Author’s response to reviews: see over
Dear Dr. Brieger:

Thank you for your consideration of our manuscript, “Asian-White Differences in Short Sleep Duration by Industry of Employment and Occupation in the US: A Cross-sectional Study,” for publication in the BMC Public Health. We are delighted to have the opportunity to provide revisions based upon the recommendations of the Journal Editorial Office. Changes in the revised manuscript to address concerns raised are described below:

Editor:

1) RE: Kindly provide the email address of all the authors on the Title Page.
We have added the email addresses of all authors to the Title Page. Thank you.
Chandra L. Jackson - cjackson@hsph.harvard.edu
Ichiro Kawachi - ikawachi@hsph.harvard.edu
Susan Redline - sredline1@rics.bwh.harvard.edu
Hee-Soon Juon - hjuon@jhsphs.edu
Frank B. Hu, MD - nhbfh@channing.harvard.edu

2) RE: Competing Interests section
Thank you. We moved added a competing interests section to the manuscript, and included the following statement: The authors declare that they have no competing interests.

We hope that the concerns raised in the review have been adequately addressed. Thank you for your consideration of this revised manuscript.

Sincerely,

Chandra L. Jackson, PhD, MS
Postdoctoral Research Fellow
Department of Nutrition
Harvard School of Public Health